Author's response to reviews

Title: Hexane fraction of Ardisia crispa Thunb A.DC root inhibits inflammation-induced angiogenesis

Authors:

Dayang Erna Zulaikha Awang Hamsin (dayangerna@hotmail.com)
Roslida Abdul Hamid (roslida@medic.upm.edu.my)
Latifah Saiful Yazan (latifah@medic.upm.edu.my)
Che Norma Mat Taib (norma@medic.upm.edu.my)
Looi Ting Yeong (joyce_ylt86@hotmail.com)

Version: 3 Date: 24 October 2012

Author's response to reviews: see over
Dear Editor,

I herewith attached the amendment (in bold) that has been done based on the comments from the reviewer.

Revisions made following review from Reviewer 1: Umesh K. K. Jain

Major Compulsory Revisions
In RESULTS: Further separation of ACRH yielded about 2.3 g (38.38% w/w) of QRF. Here the % value seems incorrect. (In our previous above remark, I would like to clarify that in manuscript amount taken of ARCH is 5 g in text and it is 6 g in Fig. It should be either 5 or 6 and so the % accordingly. Correction is made; 5 g of ACRH is corrected to 6 g of ACRH, thus the % of yield was 38.38% as in Fig.